Author's response to reviews

Title: Mortality and cause of death in hip fracture patients aged 65 or older - a population-based study

Authors:

Jorma Panula (jorma.panula@fimnet.fi)
Harri Pihlajamäki (harri.pihlajamaki@helsinki.fi)
Ville M Mattila (ville.mattila@uta.fi)
Pekka Jaatinen (pekka.jaatinen@rauma.fi)
Tero Vahlberg (tero.vahlberg@utu.fi)
Pertti Aarnio (pertti.aarnio@satshp.fi)
Sirkka-Liisa Kivelä (sirkka-liisa.kivela@utu.fi)

Version: 7 Date: 17 May 2011

Author's response to reviews:

MS: 9759436863810721 - Mortality and cause of death in hip fracture patients aged 65 or older - a population-based study

Dear Editor

Thank you very much for the acceptance of our manuscript for publication in BMC Musculoskeletal Disorders.

We have made the following formatting changes:

A functioning link to website is now provided (Reference 10).
The Title Page is now structured according to the standards of the journal.
In the Title Page: PJ's affiliation with capital letters (Rauma Health Office)
One corresponding author is indicated.
All highlightings have been removed from the text.

On behalf of our research team

Kind regards,

Jorma Panula MD PhD